Author's response to reviews

Title: CD58 polymorphisms associated with the risk of neuromyelitis optica in a Korean population

Authors:

Jason Y Kim (imblue870227@gmail.com)
Joon Seol Bae (jsbae2009@gmail.com)
Ho Jin Kim (hojinkim@ncc.re.kr)
Hyoung Doo Shin (hdshin@sogang.ac.kr)

Version: 2  Date: 19 November 2013

Author's response to reviews:

changed the format accordingly as per the guideline of BMC.